Author’s response to reviews

Title: Cardiac computed tomography imaging Follow-Up of Iatrogenic Aorto-Coronary Dunning” Dissections

Authors:

Stefan Baumann (stefan.baumann@umm.de)
Michael Behnes (m-g-n-b@web.de)
Benjamin Sartorius (benni.sartorius@yahoo.de)
Tobias Becher (tobias.becher@umm.de)
Ibrahim El-Battrawy (ibrahim.el-battrawy@umm.de)
Christian Fastner (christian.fastner@umm.de)
Uzair Ansari (uzair.ansari@umm.de)
Dirk Lossnitzer (dirk.lossnitzer@umm.de)
Kambis Mashayekhi (kambis.mashayeko@universitaets-herzzentrum.de)
Thomas Henzler (thomas.henzler@umm.de)
Stefan Schoenberg (stefan.schoenberg@umm.de)
Martin Borggrefe (martin.borggrefe@umm.de)
Ibrahim Akin (ibrahim.akin@umm.de)

Version: 3 Date: 05 Sep 2017

Author's response to reviews:

Dear Dr. Cookson,

1. Thank you very much for removing the patient age and sex information from the main text and figure legends. However, we note that these are still included in table 3 so we would be grateful if you could now remove these data. Please also amend the following sentence in the Results (page 8) to correspond with this change: "Table 3 outlines in detail each individual patient with DD, specifically regarding age, sex, target vessel, indication of the PCI."
We changed the sentence to "Table 3 outlines the cases with DD, specifically regarding target vessel and indication of the PCI."

2. Please confirm that the ethics committee approved the use of verbal consent from the participants. Please state this in the 'Ethics approval and consent to participate' section of the declarations.

We have amended the sentence to: "All patients provided informed verbal consent, which was approved by the ethics committee."

Kind regards

Stefan Baumann

Michael Behnes

Ibrahim Akin